\documentclass[11pt]{article}
\def\Z{\mathbb{Z}}
\def\Q{\mathbb{Q}}
\usepackage{latexsym,amsmath,amssymb,epsfig}
\setlength{\topmargin}{-0.5in}
\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{0in}
\setlength{\textheight}{8.0in}

\begin{document}
\pagenumbering{gobble}
\centerline{MATH 133 -- Calculus with Fundamentals 1}
\centerline{Quiz 6 -- November 3, 2017}
\vskip 40pt
\noindent
Your Name: $\underline{\qquad\qquad\qquad\qquad\qquad\qquad\qquad\qquad\qquad}$
\vskip 20pt
\noindent
{\it Directions}
\vskip 10pt
Do all work on this sheet.  There are 30 possible points.
Use the ``short-cut'' rules to find the derivatives of the following functions.
(You do not need to simplify.)
\begin{enumerate}
\item[(1)] (10) $f(x) = 6x^{8} + 2\sqrt[5]{x}  + 3\sin(x)$
\vskip 2.0truein
\item[(2)] (10) $g(x) = x^2\tan(x)$
\vskip 2.0truein
\item[(3)] (10) $\displaystyle h(x) = \frac{x + \cos(x)}{1 + e^x}$
\end{enumerate}

\end{document}
